Author's response to reviews

Title: Maternal mortality in the informal settlements of Nairobi city: What do we know?

Authors:

Abdhalah K Ziraba (akziraba@yahoo.com)
Nyovani J Madise (N.J.Madise@soton.ac.uk)
Samuel Mills (smills@ihsp.edu)
Catherine Kyobutungi (ckyobutungi@aphrc.org)
Alex Ezeh (aezeh@aphrc.org)

Version: 3 Date: 27 March 2009

Author's response to reviews:

I accepted most of the editorial changes which were in track changes as suggested by the editorial team. I also made additional typographic and grammatical changes in a few areas. Some changes in table 2 were implemented after realising that there was an error (of 4 cases) in the total number of women aged 15-49. Corresponding changes were also reflected in the text.